Author’s response to reviews

Title: Association of nutritional support programs with zinc deficiency in Colombian children: a cross-sectional study

Authors:

Angela Maria Maria Pinzon-Rondon (angela_pinzon@hotmail.com)
Alfonso Hoyos-Martínez (alfonsohoyos@gmail.com)
Daniela Parra-Correa (danielaparrac_93@hotmail.com)
Ana Maria Pedraza-Flechas (anama.pedraza@urosario.edu.co)
Angela Maria Ruiz-Sternberg (angela.ruiz@urosario.edu.co)

Version: 3 Date: 30 Jul 2019

Author’s response to reviews:

Dear Editors,

Regarding your commentaries:

1. While assessing your latest submission it was noticed that your tables were not included in your manuscript or accompanying files. Please include your tables in your main manuscript file when resubmitting.
   Answer: We have included the tables in the current submission

2. The corresponding author’s email address as displayed in our submission data and as displayed on the title page currently do not match. Please revise so that the corresponding author’s email address is consistent in both of these locations.
   Answer: The corresponding author created its profile with her personal e-mail (angela_pinzon@hotmail.com), but for the paper information we would like you to include the institutional e-mail (angela.pinzon@urosario.edu.co). We have tried to eliminated the personal email address from the submission data, but we were not able to do it.
3. When resubmitting, please include only one, final version, manuscript. Please do not include a revised reference manuscript at this point as this is not needed at this stage of the editorial process.

Answer:
We have added only the final version of the manuscript

Sincerely,

Angela Pinzon-Rondon